Author's response to reviews

Title: Comparison of m-mode echocardiographic left ventricular mass measured using digital and strip chart readings: The Atherosclerosis Risk in Communities (ARIC) Study

Authors:

Dr Donna K Arnett (arnett@epi.umn.edu)
Thomas N Skelton (tskelton@medicine.umsmed.edu)
Philip R Liebson (pliebson@rpslmc.edu)
Emelia Benjamin (emelia@bu.edu)
Richard G Hutchinson (rhutchinson@medicine.umsmed.edu)

Version: 5 Date: 6 Jun 2003

I have successfully cropped our figure to meet your needs. The manuscript remains the same as the previous final online resubmission which was uploaded 6/6/03.